
-----------------------------------------------------------------------------
          Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
       Fri Feb  9 16:49:48 GMT 2018
Year: 18  Doy: 040
Observer: berkey
WEATHER COMMENT: berkey: Fri Feb 09 16:49:47 GMT 2018
Temp: 33.5f, Humidity: 24%, Pressure: 28.668in, Wind: 14mph from 253degs, Skies: clear
___end___
GENERAL COMMENT BY berkey: Fri Feb 09 17:19:44 GMT 2018
Added a new USB ethernet controller to SGS.  Kcor can get sgs signals and we are writing out to kodiak as well.
___end___
GENERAL COMMENT BY berkey: Fri Feb 09 17:20:02 GMT 2018
Sky is bright, may not get data or good data this mroning.
___end___
Fri Feb 09 17:56:47 GMT 2018 CoMP occulter has been re-centered
GENERAL COMMENT BY berkey: Fri Feb 09 18:20:39 GMT 2018
SGS has been moved over to the main UPS
___end___
GENERAL COMMENT BY berkey: Fri Feb 09 19:04:54 GMT 2018
Wednesdays KVM issue was all user error.

After troubleshooting the ethernet issue, I plugged the SGS video output into the KVM extender video output instead of inp
ut.  Today with the video cable connected to the correct plug it works agian.
___end___
Fri Feb 09 19:48:53 GMT 2018 SGS Alignment complete
Fri Feb 09 19:49:35 GMT 2018 CoMP occulter has been re-centered
Fri Feb 09 19:49:40 GMT 2018 COMP Start Patrol on cookbook: synoptic-00012.cbk
Fri Feb 09 19:49:53 GMT 2018 Kcor Focus/alignment program exited
Fri Feb 09 21:24:21 GMT 2018 COMP End Patrol
Fri Feb 09 21:24:22 GMT 2018 COMP Start Patrol on cookbook: synoptic-00012.cbk
Fri Feb 09 22:58:55 GMT 2018 COMP End Patrol
Fri Feb 09 22:58:56 GMT 2018 COMP Start Patrol on cookbook: waves-00003.cbk
Fri Feb 09 23:28:29 GMT 2018 SGS Alignment complete
Sat Feb 10 00:32:02 GMT 2018 COMP End Patrol
Sat Feb 10 00:32:03 GMT 2018 COMP Start Patrol on cookbook: synoptic-00012.cbk
Sat Feb 10 02:06:41 GMT 2018 COMP End Patrol
Sat Feb 10 02:06:42 GMT 2018 COMP Start Patrol on cookbook: synoptic-00012.cbk
Sat Feb 10 02:06:47 GMT 2018 COMP End Patrol
GONG COMMENT BY berkey: Sat Feb 10 02:12:09 GMT 2018
Swapped the gong INST ADC board today. But this didn’t fix the problem.
___end___
ONSITE STAFF:  berkey


